Author’s response to reviews

Title: Acute and mixed alcohol intoxications in asylum seekers presenting to an urban emergency department in Switzerland

Authors:

Adam Brown (brownad@newschool.edu)

Martin Müller (martin.mueller2@insel.ch)

Trevor Hirschi (trevor.hirschi@insel.ch)

Jonathan Henssler (jonathan.henssler@charite.de)

Katharina Rönz (katharina.roenz@insel.ch)

Aristomenis Exadaktylos (aristomenis.exadaktylos@insel.ch)

David Srivastava (david.srivastava@insel.ch)

Version: 3 Date: 26 Apr 2019

Author’s response to reviews:

Editor Comments:

1. Please ensure that all tables and figures are included in your submission when uploading your manuscript. We note that currently the tables and fig.1 are missing.

Thank you for noting this. It had been uploaded to the portal, but it has been uploaded again.

Thank you very much.

Best,

Adam